Author's response to reviews

Title: Do acute elevations of serum creatinine in primary care engender an increased mortality risk?

Authors:

Helen Hobbs (helen.hobbs@nhs.net)
Paul Bassett (paul@statsconsultancy.co.uk)
Toby Wheeler (t wheeler@nhs.net)
Michael Bedford (michael.bedford@nhs.net)
Jean Irving (jirving2@nhs.net)
Paul E Stevens (p stevens@nhs.net)
Christopher K Farmer (chris.farmer1@nhs.net)

Version: 4 Date: 1 December 2014

Author's response to reviews:

Dear Editor,

Thank you for accepting our article. Please find responses to the editorial requests:

1. Kindly ensure that the final Kaplan-Meier graph is redrawn? (Comment 5, 1st Reviewer)
   These have been redrawn
2. Title page? It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
   This has been updated
3. Formatting - You now have an opportunity to check that your revised manuscript conforms to the journal style
   The formatting complies with the journal style

Many thanks for publishing this article in BMC Nephrology

Yours Sincerely

Chris Farmer